Author's response to reviews

Title: Bezoar in gastro-jejunosotomy presenting as gastric outlet obstruction: case report and review of the literature

Authors:

   Edmund Leung (ed.leung@doctors.org.uk)
   Ruth Barnes (ruth.barnes@hotmail.com)
   Ling S Wong (lingswong@doctors.org.uk)

Version: 2 Date: 29 March 2008

Author's response to reviews: see over
Dear Editor,

Bezoar in gastro-jejunostomy presenting as gastric outlet obstruction: case report and review of the literature.

Edmund Leung, Ruth Barnes, Ling Wong

Please find attached submission of a case report. The authors feel that there is clinical importance of the article, not only for its unusual presentation; but also for the complication presented itself over 20 years after the patient’s surgery performed in Kenya.

I hope you will find the article interesting and relevant.

Yours sincerely,

Dr Edmund Leung